Author’s response to reviews

Title: ‘Mankind owes to the child the best that it has to give’: Prison conditions and the health situation and rights of children incarcerated with their mothers in sub-Saharan African prisons.

Authors:

Marie Claire Van Hout (m.c.vanhout@ljmu.ac.uk)
Rosemary Mhlanga-Gunda (rosemhlanga3@gmail.com)

Version: 1 Date: 22 Oct 2018

Author’s response to reviews:

Dear Editor,

Many thanks for the two suggested changes. I have actually removed the term circumstantial and revised accordingly to reflect ‘incarcerated children with their mothers in prison’. I have also amended Sierra Leone throughout the paper. It would not however give me the option of changing the title on your system. It is correct in the uploaded manuscript.

Kind Regards

Prof. Marie Claire Van Hout